\documentclass[11pt]{amsart}
\usepackage{graphicx, color}
\usepackage{amscd}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{mathrsfs}
\textwidth=6in \textheight=8in \topmargin=-0.5cm
\oddsidemargin=0.5cm \evensidemargin=0.5cm
%\usepackage[notref,notcite]{showkeys}

\newtheorem{theorem}{Theorem}
\newtheorem{ex}[theorem]{Example}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{claim}[theorem]{Claim}
\newtheorem{step}[theorem]{Step}
\newtheorem*{theorem*}{Theorem}
\newenvironment{proofcor1}{\emph{Proof of Corollary~\ref{corollarylsc}:}}
{\hspace*{\fill}\mbox{$\Box$}}
\newenvironment{proofcor2}{\emph{Proof of Corollary~\ref{thesistenzah}:}}
{\hspace*{\fill}\mbox{$\Box$}}
\newenvironment{proofcor3}{\emph{Proof of Corollary~\ref{corollary}:}}
{\hspace*{\fill}\mbox{$\Box$}}

\numberwithin{equation}{section}
\newcommand{\RR}{\mathbb R}
\newcommand{\R}{\mathbb R}
\newcommand{\erre}{\mathbb R}
\newcommand{\NN}{\mathbb N}
\newcommand{\PP}{\mathbb P}
\newcommand{\dist}{\texttt{\rmfamily{dist}}}
\renewcommand{\le}{\leqslant}
\renewcommand{\leq}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\geq}{\geqslant}
\def\di{\displaystyle}
\def\ri{\rightarrow}
\def\meas{\mathop{\rm meas}}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\baselineskip=16pt plus 1pt minus 1pt

\begin{document}
\begin{thebibliography}{99}

\bibitem{armitage} D.H. Armitage and S.J. Gardiner, {\it Classical Potential Theory}, Springer, London, 2000.

\bibitem{axler} S. Axler, P. Bourdon, and W. Ramey, {\it Harmonic Function Theory}, Graduate Texts in Mathematics, vol. 137, Springer, New York, 2001.

\bibitem{brekam}  H. Brezis and S. Kamin,  Sublinear elliptic equations in ${\mathbb R}^n$, {\it Manuscripta Math.} {\bf 74} (1992), 87-106.

\bibitem{crjmaa} F. C\^{\i}rstea and V. R\u adulescu,  Existence and uniqueness of positive solutions to a semilinear elliptic problem in $\RR^N$, {\it J. Math. Anal. Appl.} {\bf 229} (1999), 417-425.

\bibitem{crna} F. C\^{\i}rstea and V. R\u adulescu, Blow-up solutions for semilinear elliptic problems, {\it Nonlinear Anal.} {\bf 48} (2002), 541-554.

\bibitem{crccm} F. C\^{\i}rstea and V. R\u adulescu, Existence and uniqueness of blow-up solutions for a class of logistic equations, {\it Commun. Contemp. Math.} {\bf 4} (2002), 559-586.

    \bibitem{graaa} M. Ghergu and V. R\u adulescu, Existence and nonexistence of entire solutions to the logistic differential equation, {\it Abstract and Applied Analysis} {\bf 17} (2003), 995-1003.

    \bibitem{grcpaa} M. Ghergu and V. R\u adulescu, Nonradial blow-up solutions of sublinear elliptic equations with gradient term, {\it Communications on Pure and Applied Analysis} {\bf 3} (2004), 465-474.

    \bibitem{lair} A.V. Lair and A.W. Wood, Large solutions ofsublinear elliptic equations, {\it Nonlinear Anal.} {\bf 39} (2000), 745-753.

\bibitem{el1} K. El Mabrouk, Entire bounded solutions for a class of sublinear elliptic equations, {\it Nonlinear Anal.} {\bf 58} (2004),  205-218.

%\bibitem{el2} K. El Mabrouk and W. Hansen,  Nonradial large solutions of sublinear elliptic %problems, {\it J. Math. Anal. Appl.} {\bf 330} (2007), 1025-1041.

\bibitem{psjde} P. Pucci and J. Serrin, The strong maximum principle revisited, {\it J. Differential Equations} {\bf 196} (2004), 1-66.

\bibitem{pucser} P. Pucci and J. Serrin, {\it The Maximum Principle}, Progress in Nonlinear Differential Equations and their Applications, vol.~73,  Birkh\"auser Verlag, Basel, 2007.

\bibitem{handbook} V. R\u adulescu, Singular phenomena in nonlinear elliptic problems. From blow-up boundary solutions to equations with singular nonlinearities, in {\it Handbook of Differential Equations: Stationary Partial Differential Equations}, Vol. 4 (M.~Chipot, Ed.), North-Holland Elsevier Science, Amsterdam, 2007, pp. 483-591.

\bibitem{cluj1}  A. Petru\c sel, D.R. Sahu, and J.C. Yao,  On fixed points of pointwise Lipschitzian type mappings, {\it Fixed Point Theory} {\bf 14} (2013), 171-184.

\bibitem{cluj2} I.A. Rus, A. Petru\c sel, and G. Petru\c sel, {\it Fixed Point Theory}, Cluj University Press, Cluj-Napoca, 2008.

\bibitem{cluj3} I.A. Rus, An abstract point of view on iterative approximation of fixed points: impact on the theory of fixed point equations, {\it Fixed Point Theory} {\bf 13} (2012),  179-192.
\end{thebibliography}

%\hfill\today\bigskip

\title[Nonlinear elliptic equations with variable potential]{Entire bounded solutions versus fixed points for nonlinear elliptic equations with indefinite weight}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\author[R. Alsaedi]{Ramzi Alsaedi}
\address{Department of Mathematics, Faculty of Sciences, King Abdulaziz University,
P.O. Box 80203, Jeddah 21589, Saudi Arabia}
\email{\tt ramzialsaedi@yahoo.co.uk}

\author[H. M\^aagli]{Habib M\^aagli}
\address{Department of Mathematics, Faculty of Sciences, King Abdulaziz University,
P.O. Box 80203, Jeddah 21589, Saudi Arabia}
\email{\tt abobaker@kau.edu.sa}

\author[V.D. R\u{a}dulescu]{Vicen\c{t}iu D. R\u{a}dulescu}
\address{Department of Mathematics, Faculty of Sciences, King Abdulaziz University,
P.O. Box 80203, Jeddah 21589, Saudi Arabia}\email{vicentiu.radulescu@imar.ro}

\author[N. Zeddini]{Noureddine Zeddini}
\address{Department of Mathematics, Faculty of Sciences, King Abdulaziz University,
P.O. Box 80203, Jeddah 21589, Saudi Arabia}
\email{\tt noureddine.zeddini@ipein.rnu.tn}

\begin{abstract} We establish a necessary and sufficient condition for the existence of an entire distributional solution for a general class of nonlinear elliptic equations with variable potential and nondecreasing nonlinear term. Our result establishes the relationship between the Green function and the growths of the weight and of the nonlinear term. The main result also points out the connection with a fixed point problem for an integral operator.
\\
\ \\
{\sc Key words}: nonlinear elliptic equation, entire solution, fixed point, Green function.\\
\ \\
{\sc 2010 AMS Subject Classifications}. Primary: 35B50, 35J61.
Secondary: 35J67, 35J75, 58J32.
\end{abstract}

\maketitle

\tableofcontents

\section{Introduction}\label{sec1}
Nonlinear elliptic problems on the whole space do not have necessarily a solution. The existence of solutions is in relationship not only with the nonlinearity but also with the growth of the variable potential involved in the problem.

In their seminal paper on entire solutions of sublinear elliptic equations, Brezis and Kamin \cite{brekam} pointed out a striking phenomenon. They showed that a  sublinear elliptic problem on the whole space has a solution if and only if a related {\it linear} partial differential equation depending only on the potential has a solution. Brezis and Kamin considered the nonlinear problem
\begin{equation} \label{bre1}
-\Delta u = \rho (x)\, u^\alpha,\quad x\in
\mathbb{R}^N\ (N\geq 3),
\end{equation}
with $0<\alpha<1$, $\rho\in L^\infty_{\rm loc}(\mathbb{R}^N)\setminus\{0\}$, $\rho\geq 0$.

The main result in \cite{brekam} establishes that problem \eqref{bre1} has a bounded positive solution if and only if the linear equation
$$-\Delta u = \rho (x),\quad x\in
\mathbb{R}^N$$
has a bounded solution, namely if the mapping
$$\RR^N\ni x\longmapsto \int_{\RR^N}\frac{\rho (y)}{|x-y|^{N-2}}\,dy$$
is in $L^\infty (\RR^N)$. This result points out the relationship between the growth of the potential and the Green function on the whole space.


 The analysis developed in \cite{brekam} shows that a bounded solution of problem
 \eqref{bre1}
 exists for potentials like
$$\rho(x)=\frac{1}{1+|x|^q}\quad\mbox{or}\quad \rho(x)=\frac{1}{(1+|x|^2)\,|\log (2+|x|)|^q}\ \ (q>2),$$
while no solution exists if
$$\rho(x)=\frac{1}{1+|x|^q}\quad\mbox{with $q\leq 2$}\,.$$

Brezis and Kamin \cite{brekam} proved also that a stronger nonexistence result holds, provided that
$$\int_{|x|\geq 1}\frac{\rho(x)}{|x|^{N-2}}dx=+\infty .$$
In such a case there is no function $u\in L^1_{\rm loc}(\RR^N)$ satisfying
$$\left\{
\begin{array}{lll}
&\di -\Delta u=\rho (x)\, u^\alpha&\quad \mbox{in}\
\mathbb{R}^N\\
& u\geq 0&\quad \mbox{in}\
\mathbb{R}^N.\end{array}\right.
$$

In the present paper, we are concerned with a related problem on the whole space.
The main novelties in our approach are the following:

(a) We study a class of nonlinear elliptic equations with general nonlinear term, which is not only of power-type.

(b) We consider a reversed sign for the variable potential with respect to the framework considered by Brezis and Kamin \cite{brekam}. Due to this assumption, we are not concerned with super-harmonic functions, and not with subharmonic functions defined on the whole Euclidean space.

(c) The methods that we develop in the present paper can be applied to larger classes of differential operators. Indeed, our proofs remain valid in the general setting of $A$-Laplace operators formulated by Pucci and Serrin \cite{psjde, pucser}, namely differential operators of the type
\begin{equation}\label{pucci}\mbox{div}\,(A(|\nabla u|)\nabla u),\end{equation}
where $A=A(\rho)\in C(0,\infty)$ and
$$\rho\longmapsto \rho A(\rho)\ \mbox{is strictly increasing in}\ (0,\infty)\ \mbox{and}\
\rho A(\rho)\ri 0\ \mbox{as}\ \rho\ri 0.$$
This framework includes the degenerate $p$-Laplace operator ($A(\rho)=\rho^{p-2}$, $p>1$) and the mean curvature operator ($A(\rho)=1/\sqrt{1+\rho^2})$.

(d) Our necessary and sufficient condition for the existence of an entire distributional solution is established in terms of the growth of the variable potential and the nonlinear term, in relationship with the Green function on the whole space. The entire solution appears as the fixed point of a suitable integral operator.
%for instance the $p$-Laplace operator, the mean curvature operator
%$$\mbox{div}\,\left(\frac{\nabla u}{\sqrt{1+|\nabla u|^2}}\right),$$
%but also to operators like
%$$\mbox{div}\,\left((1+|\nabla u|^2)^{-\alpha}\nabla u \right)\quad (\alpha\leq -\frac12)$$
%or $$\mbox{div}\,\left(|\nabla u|^{2p-2}(1+|\nabla u|^{2p})^{-1/2}\,\nabla u\right)\quad (p\geq %1).$$

Our paper extends some recent results from  \cite{el1}, where it is studied only the sublinear case corresponding to $f(u)=u^\rho$ with $\rho\in (0,1)$. The analysis carried out in the present paper allows to extend the results in \cite{el1} to larger classes of nonlinearities, including to those with linear behaviour like
$$f(u)=\frac{u}{\log (u+1)},\quad f(u)=\frac{u}{\sqrt{u^2+1}}\quad\mbox{or}\quad f(u)=\frac{\arctan u}{u}\,.$$
We also point out that general classes of nonlinear elliptic problems on the whole space can be treated by means of our results, including problems with superlinear terms. The monotonicity assumption of the nonlinear term is necessary only in some cases. As in \cite{lair} and \cite{el1}, our main result establishes a necessary and sufficient condition for the existence of bounded solutions in terms of the relationship between the Green function and the growths of the indefinite potential and the nonlinear term. This idea is used in the previous papers \cite{crjmaa}
\cite{crna}, \cite{crccm}, \cite{graaa}, \cite{grcpaa}, and  \cite{handbook}.
We also refer to the works \cite{cluj1}, \cite{cluj2}, \cite{cluj3} for a relationship between solutions of nonlinear partial differential equations and fixed points of related nonlinear operators.

Consider the nonlinear problem
\begin{equation}\label{p1}\left\{\begin{array}{lll}
&\Delta u = V (x)\, f(u)&\qquad \mbox{in}\
\mathbb{R}^N\ (N\geq 3)\\
& u\geq 0,\ u\not=0&\qquad \mbox{in}\
\mathbb{R}^N\,, \end{array}\right.
\end{equation}
where $V\geq 0$ is a nontrivial weight and $f\geq 0$ is nondecreasing.
%The model example here is $f(u)=u^p$ ($p>1$) but nonlinear terms with a different structure can be %involved in our analysis, for instance
%$$f(u)=\frac{u}{\log (u+1)},\quad f(u)=\frac{u}{\sqrt{u^2+1}}\quad\mbox{or}\quad f(u)=\frac{\arctan %u}{u}\,.$$

We point out that problem \eqref{p1} is not of interest if $N\in\{1,2\}$. Indeed, the solution should be subharmonic and nonnegative. Thus by the Liouville theorem, $u$ is constant provided that $N\in\{1,2\}$, which attracts $V=0$ in $\RR^N$.

Throughout this paper we assume that $V\in L^\infty_{\rm loc}(\mathbb{R}^N)\setminus\{0\}$, $V\geq 0$, and $f:\RR\ri\RR$ is a nondecreasing Borel function.

We are interested in distributional solutions of problem \eqref{p1}, namely solutions in ${\mathcal D}'(\RR^N)$. Thus we say that $u:\RR\ri\RR$ is a solution of problem \eqref{p1} if $u$ is continuous, $$\mbox{the mapping $\RR^N\ni x \longmapsto V(x)f(u(x))$ is in $L^1_{\rm loc}(\mathbb{R}^N)$},$$ and for all $v\in C^2_{c}(\RR^N)$
$$\int_{\RR^N}u\Delta vdx=\int_{\RR^N}V(x)f(u)vdx.$$

\section{Case of bounded domains}
The basic idea to solve problem \eqref{p1} is to approximate it with equations on bounded domains of higher and higher measure. The most natural is to consider open balls $B(0,n)$ in  $\RR^N$ and then to take $n\ri\infty$. We perform our analysis on balls of the Euclidean space even if similar existence results hold for more general open bounded sets $\Omega\subset \RR^N$. We refer to {\it regular} domains $\Omega$, which are characterized by the property that  all continuous function $g:\partial\Omega\ri\RR$ admits a harmonic extension $H(g;\Omega)$ to $\Omega$. In other words, these are domains $\Omega$ for which the problem
$$\left\{\begin{array}{lll}
&\Delta H(g;\Omega)=0&\quad\mbox{in}\ \Omega\\
& H(g;\Omega)=g&\quad\mbox{on}\ \partial\Omega\,
\end{array}\right.
$$
has a solution.

We recall that a function $u:\Omega\ri\RR$ is super-harmonic if the following properties hold:

(i) $u$ is lower-semicontinuous;

(ii) $H(u;\omega)\leq u$ in $\omega$, for all $\omega\subset\subset\Omega$.

The function $u:\Omega\ri\RR$ is sub-harmonic if $-u$ is  super-harmonic.



\subsection{Formulation of the problem}
Throughout this section we assume the following hypotheses:
\begin{equation}\label{v1}
V\in L^\infty_{\rm loc}(\mathbb{R}^N)\setminus\{0\}\ \mbox{and}\ V\geq 0\ \mbox{in}\ \RR^N;
\end{equation}
\begin{equation}\label{f1}
f:[0,+\infty) \ri [0,+\infty)\ \mbox{is Borel measurable, nondecreasing, nontrivial, and $f(0)=0$}.
\end{equation}

Let $\Omega\subset\RR^N$ be a regular domain. We are interested in distributional solutions of the nonlinear problem
\begin{equation}\label{p1b}\left\{\begin{array}{lll}
&\Delta u = V (x)\, f(u)&\qquad \mbox{in}\
{\mathcal D}'(\Omega)\\
& u\geq 0,\ u\not=0&\qquad \mbox{in}\
\Omega\,. \end{array}\right.
\end{equation}
This means that $u:\Omega\ri [0,+\infty)$ is a continuous nontrivial function such that $Vf(u)\in L^1_{\rm loc}(\Omega)$ and
for all $v\in C^2_{c}(\Omega)$
$$\int_{\Omega}u\Delta vdx=\int_{\Omega}V(x)f(u)vdx.$$

Let $\Omega\subset\RR^N$ be a ball (or even a regular bounded domain) such that $V\not=0$ in $\Omega$. Assume that
\begin{equation}\label{g1}
g : \overline\Omega\ri [0,+\infty)\ \mbox{is a continuous notrivial function}.
\end{equation}

Consider the nonlinear elliptic problem
\begin{equation}\label{p1marg}\left\{\begin{array}{lll}
&\Delta u = V (x)\, f(u)&\qquad \mbox{in}\
{\mathcal D}'(\Omega)\\
& u=g&\qquad \mbox{on}\
\partial\Omega\\
& u\geq 0,\ u\not=0&\qquad \mbox{in}\
\Omega .\end{array}\right.
\end{equation}

We say that $\underline u:\overline\Omega\ri\RR$ is a lower-solution of
\begin{equation}\label{lower}\Delta u = V (x)\, f(u)\qquad \mbox{in}\
{\mathcal D}'(\Omega)\end{equation} if
$\underline u$ is continuous, $Vf(\underline u)\in L^1_{\rm loc}(\Omega)$, and
for all $v\in C^2_{c}(\Omega)$ with $v\geq 0$
$$\int_{\Omega}u\Delta vdx\geq\int_{\Omega}V(x)f(u)vdx.$$

The function $\overline u:\Omega\ri\RR$ is an upper-solution of  problem \eqref{lower} if $-\overline u$ is a lower-solution of this problem.

The weak comparison principle asserts that if $\underline u$ (resp., $\overline u$) is lower-solution (resp., upper-solution) of problem \eqref{lower} such that
\eqref{f1} holds and
$$\liminf_{x\ri y}(\overline u-\underline u)(x)\geq 0\quad\mbox{for all $y\in\partial\Omega$},$$
then $\underline u\leq \overline u$ in $\omega$. We refer to Pucci and Serrin \cite{pucser} for details (see also C\^{\i}rstea and R\u adulescu \cite{crccm}).

\subsection{The main existence result}
In this section, we prove an existence and uniqueness property, which extends Theorem 5 in C\^{\i}rstea and  R\u adulescu \cite{crna} and Theorem A.1 in C\^{\i}rstea and R\u adulescu \cite{crccm}. We state the main result in the framework of balls, even if it still holds for bounded regular domains $\Omega\subset\RR^N$ that have a Green function.

Let $G$ be the Green function of $\Omega$. Then for all $y\in\Omega$
$$-\Delta G(\cdot ,y)=\delta_y\quad\mbox{in}\ \Omega,$$
where $\delta_y$ is the Dirac mass at $y$.

If $u:\Omega\ri\RR$ is a Borel measurable function, we define the integral operator
$$G_\Omega u(x):=\int_\Omega G(x,y)u(y)dy$$
for all $x\in\Omega$ such that the integral makes sense.

Cf. Armitage and Gardiner \cite{armitage} for all Borel function $u\geq 0$, the following alternative holds:

(i) either $G_\Omega u(x)=+\infty$ for all $x\in\Omega$\\
or

(ii) 0 is the only nonnegative harmonic function bounded above by $G_\Omega u$ in $\Omega$.

The operator $G_\Omega$ has the following additional properties:

($G_1$) $G_\Omega$ is compact as defined on the space of bounded Borel functions;

($G_2$) $G_\Omega u$ is a bounded Borel function, provided that $u$ is bounded. Moreover,
$$\lim_{x\ri y}G_\Omega u(x)=0\quad\mbox{for all}\ y\in\partial\Omega .$$

This operator will play a central role in the construction of the solution of problem \eqref{p1marg}. This solution will be obtained as the fixed point of an operator that involves both $G_\Omega$ and the harmonic extension operator.

\begin{theorem}\label{t1}
Let $\Omega\subset\RR^N$ be a ball. Assume that hypotheses
\eqref{v1}, \eqref{f1}, \eqref{g1} are fulfilled, and $V\not=0$ in $\Omega$. Then problem \eqref{p1marg} has a unique distributional solution.
\end{theorem}

{\it Proof.} Let us first show that if a solution exists then it must be unique. Assume that
$u_1$ and $u_2$ are distributional solutions of problem \eqref{p1marg} and denote
$$\omega:=\{x\in\Omega;\ u_1(x)<u_2(x)\}\,.$$
Set $u:=u_1-u_2$. Assuming by contradiction that $\omega\not=\emptyset$ we have $u>0$ in $\omega$ and
$$\Delta u=V(x)(f(u_1)-f(u_2))\leq 0\quad\mbox{in}\ {\mathcal D}'(\omega).$$
Thus $u$ is a superharmonic function in $\omega$ satisfying
$$\liminf_{x\ri y}u(x)\geq 0\quad\mbox{for all $y\in\partial\Omega$}.$$
Using  the maximum principle, we obtain $u\geq 0$. This contradicts the definition of $\omega$. We deduce that $\omega=\emptyset$, hence $u_1\geq u_2$ in $\Omega$. A similar argument shows that
$u_1\leq u_2$ in $\Omega$. This shows that $u_1=u_2$.

In order to establish the existence of the solution, we see that 0 is a lower-solution and $n$ is an upper-solution (for $n$ large enough). Thus a solution of problem \eqref{p1marg} exists
but we cannot guarantee that it is nontrivial. In fact, as remarked by Pucci and Serrin \cite{pucser}, the {\it dead core phenomenon} can occur, namely the solution exists, is nonnegative but it vanishes in an interior region. This is characteristic to the  sublinear framework, which is including in the setting of the present paper. In such a case, the maximum principle does not hold. In fact, as established by Pucci and Serrin \cite{psjde, pucser}, the maximum principle holds for operators like in \eqref{pucci} if and only if $f>0$ on some neighbourhood $(0,\delta)$ and
$$\int_0^\delta\frac{ds}{H^{-1}(F(s))}=+\infty ,$$
where $F(s):=\int_0^tf(t)dt$ and
$$H(\rho):=\rho^2A(\rho)-\int_0^\rho sA(s)ds,\qquad\rho\geq 0.$$

The following auxiliary result extends Lemma 1 in El Mabrouk \cite{el1} to more general nonlinearities. This result asserts that solutions of problem \eqref{p1marg} are fixed points of the integral operator
$$Tu(x)=H(g;\omega)(x)-\int_\Omega G(x,y)V(y)f(u(y))dy$$
defined on the space of locally bounded nonnegative Borel functions.

\begin{lemma}\label{l1}
Let $u:\Omega\ri\RR$ be a locally bounded nonnegative Borel function. Then $u$ is a solution of problem \eqref{p1marg} if and only if for all open regular subset $\omega\subset\subset\Omega$
$$u(x)+\int_\omega G(x,y)V(y)f(u(y))dy=H(g;\omega)(x)\quad\mbox{for all}\ x\in\omega.$$
\end{lemma}

{\it Proof of Lemma \ref{l1}}. By the uniqueness of the harmonic extension, it is enough to
show that the function
\begin{equation}\label{harmo}
v( x):= u(x)+\int_\omega G(x,y)V(y)f(u(y))dy\quad\mbox{for all}\ x\in\omega
\end{equation}
is harmonic, for all open regular set $\omega\subset\subset\Omega$.
Fix such a subset $\omega$.
 We first observe that since $u$ is bounded, then $v$ is a bounded function. Using now the definition of $v$ we deduce that $v$ is continuous if and only if $u$ is continuous. Fix $\varphi\in C^2_{\rm loc}(\omega)$. Multiplying by $\Delta\varphi (x)$ in the expression of $v$ and integrating in $\omega$, we find
 $$\int_\omega v\Delta\varphi dx=\int_\omega u\Delta\varphi dx-\int_\omega V(x)f(u(x))\varphi dx.$$
 This relation shows that $v$ is harmonic in $\omega$ if and only if
 $u$ is a solution of problem \eqref{p1marg}. The proof of Lemma \ref{l1} is concluded.\qed

 \smallskip
 Returning to the proof of Theorem \ref{t1}, set $M:=\|g\|_{L^\infty (\partial\Omega)}$ and define the bounded function $\zeta:\RR\ri\RR$ by
 $$\zeta (s)=\left\{\begin{array}{lll}
 & \min\{f(s),f(M)\}&\quad\mbox{if} \ s\geq 0\\
 & 0 &\quad\mbox{if} \ s< 0.
 \end{array}\right.
 $$

 Define the functions $u_n:\omega \ri\RR$ by $u_0=0$ and
 $$u_n=H(g;\omega)-z_n\quad\mbox{for all}\ n\geq 1,$$
 where
 $$z_n(x):=G_\omega (V\zeta(u_{n-1}))(x)=\int_\omega G (x,y)V(y)\zeta (u_{n-1}(y))dy\quad\mbox{for all}\ n\geq 1.$$

 Since $\zeta\in L^\infty (\RR)$, we obtain that $(z_n)$ is a bounded sequence. Using the fact that the operator $G_\omega$ is compact, we can assume up to a subsequence that $(z_n)$ converges in $L^\infty(\Omega)$ to $u:\omega\ri\RR$. Taking now $n\ri\infty$ in the definition of $u_n$ we obtain
 $$u(x)=H(g;\omega)(x)-\int_\omega G(x,y)V(y)f(u(y))dy,$$
 for all open regular set $\omega\subset\subset\Omega$. By Lemma \ref{l1} we obtain that $u$ is a distributional solution of problem \eqref{p1marg}.
\qed

\smallskip
The hypothesis that $f$ is nondecreasing has been crucial to prove the uniqueness of the solution in Theorem \ref{t1}. This result establishes the existence of a unique {\it nonnegative} solution. In view of the compact support property related to the maximum principle (see Pucci and Serrin \cite{pucser}), the fact that this solution is {\it positive} strongly depends on the growth of the nonlinear term $f$. In fact, if $f$ has a superlinear behaviour (say, $f(u)=u^p$ with $p>1$) then the unique solution of problem \eqref{p1marg} is {\it positive}. If $f$ has a sublinear decay (say, $f(u)=u^p$ with $0<p<1$) then this solution is nonnegative but not necessarily positive. Indeed, let us consider $\Omega=B(0,1)\subset \RR^N$  $(N\geq 3)$, $f(u)=u^{\gamma/(\gamma+2)}\ (\gamma>0)$, $ V=1$, and $$g=\left[(\gamma+2)^2+(N-2)(\gamma+2) \right]^{-(\gamma+2)/2}.$$
Then problem \eqref{p1marg} admits the unique solution $u(x)=g\,|x|^{\gamma+2}\geq 0$, which vanishes at the origin.

\section{Case of the whole space}
In this section, we analyze the existence of a solution of problem \eqref{p1}. This is performed in relationship with the growth of the potential $V$ combined with the Green function of the whole space. As in \cite{crna}, the main idea is to approximate problem \eqref{p1} with problems on balls $B(0,n)$ and then to study the asymptotic behaviour as $n\ri\infty$.

We start with an useful result, which extends to our subharmonic setting a classical property pf harmonic functions.
The Baire theorem implies that if $u$ is the pointwise limit of a sequence of harmonic functions on $\Omega$, then $u$ is harmonic on a dense subset of $\Omega$. The following result establishes a related property for a sequence of nonnegative solutions of problem \eqref{p1marg}.
The proof makes use of the characterization property established in Lemma \ref{l1}.

\begin{lemma}\label{l2}
Assume that $u$ is the pointwise limit of a sequence of nonnegative, locally uniformly bounded solutions of equation \eqref{p1marg}. Then $u$ is also a solution of equation \eqref{p1marg}.
\end{lemma}

{\it Proof}.
Let $\omega\subset\subset\Omega$ an arbitrary open regular set. Applying Lemma \ref{l1} we obtain for all $x\in\omega$
\begin{equation}\label{relrel}
u_n(x)+\int_\omega G(x,y)V(y)f(u_n(y))dy=H(u_n|_{\partial\Omega};\omega)(x).
\end{equation}
Set $v_n:=H(u_n|_{\partial\Omega};\omega)$. Then $(v_n)$ is a bounded sequence of harmonic functions on $\omega$, which converges pointwise. Thus $(v_n)$ converges to a harmonic function.
Passing to the limit as $n\ri\infty$ in relation \eqref{relrel} we obtain for all $x\in\omega$
$$u(x)+\int_\omega G(x,y)V(y)f(u(y))dy=H(u|_{\partial\Omega};\omega)(x),$$
hence is a solution of problem \eqref{p1marg}.
\qed

\smallskip
Consider the nonlinear elliptic problem
\begin{equation}\label{p1n}\left\{\begin{array}{lll}
&\Delta u = V (x)\, f(u)&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
& u=1&\qquad \mbox{if}\
|x|=n\\
& u\geq 0&\qquad  \mbox{in}\
B(0,n) .\end{array}\right.
\end{equation}

As established in Theorem \ref{t1}, problem \eqref{p1n} has a unique solution $u_n$. By the maximum principle we have $u_n\leq 1$. Set
$$\tilde u_n(x):=\left\{\begin{array}{lll}
&  u_n(x)&\qquad\mbox{if}\ |x|\leq n\\
& 1 &\qquad\mbox{if}\ |x|>n.\end{array}\right.
$$
Using again the maximum principle we deduce that $(\tilde u_n)$ is a nonincreasing sequence of nonnegative functions.

\begin{proposition}\label{propo1}
Let
\begin{equation}
\label{udefinition}
u(x):=\inf\{ \tilde u_n(x);\ n\geq 1\}.\end{equation}
Then the following properties hold.

(i) We have
\begin{equation}\label{usolves}
\Delta u =V(x)f(u)\quad\mbox{in}\ {\mathcal D}'(\RR^N).
\end{equation}

(ii) The following alternative is true: either

(a) $u=0$\\
or

(b) $\sup_{x\in\RR^N}u(x)=1$ and for all $x\in\RR^N$
\begin{equation}\label{enunt}
u(x)+\int_{\RR^N}G(x,y)V(y)f(u(y))dy=1.\end{equation}
\end{proposition}

{\it Proof.}
(i) By definition, $u$ is the pointwise limit of a sequence $(u_n)$ of nonnegative bounded functions satisfying equation \eqref{lower}. Applying Lemma \ref{l2}
we deduce that $u$ solves the same equation.

(ii) If $\sup_{x\in\RR^N}u(x)=0$ then assertion (a) holds.

Assume that $\sup_{x\in\RR^N}u(x)=M\in (0,1]$. We first deduce that $M=1$. For this purpose it is enough to show that
\begin{equation}\label{cui} u\leq Mu_n\quad\mbox{for all}\ n\geq 1.\end{equation}

Consider the boundary value problems
$$\left\{\begin{array}{lll}
&\Delta v = V (x)\, f(v)&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
&\di v=\frac uM &\qquad \mbox{if}\
|x|=n\\
& v\geq 0&\qquad  \mbox{in}\
B(0,n) \end{array}\right.
$$
and
$$\left\{\begin{array}{lll}
&\Delta w = V (x)\, f(w)&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
&\di w=u &\qquad \mbox{if}\
|x|=n\\
& w\geq 0&\qquad  \mbox{in}\
B(0,n). \end{array}\right.
$$

By Theorem \ref{t1}, these problems have unique solution. We first obtain
$$\left\{\begin{array}{lll}
&\Delta (v-u_n) = V (x)\, (f(v)-f(u_n))&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
&\di v-u_n\leq 0 &\qquad \mbox{if}\
|x|=n. \end{array}\right.
$$
Since $f$ is increasing we deduce by the maximum principle that
\begin{equation}\label{eqo1} v\leq u_n\quad\mbox{in}\ B(0,n).\end{equation}

On the other side we have
$$\left\{\begin{array}{lll}
&\Delta (Mv-w) \leq V (x)\, (f(Mv)-f(w))&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
&\di Mv-w= 0 &\qquad \mbox{if}\
|x|=n. \end{array}\right.
$$
Using again the maximum principle in combination with the information that $M\in (0,1]$ we find
\begin{equation}\label{eqo2} Mv\geq w \quad\mbox{in}\ B(0,n).\end{equation}

Combining relations \eqref{eqo1} and \eqref{eqo2}, we obtain \eqref{cui}.

It remains to prove relation \eqref{enunt}. Since $u$ solves \eqref{usolves}, we find by Lemma \ref{l1} that for all $x\in B(0,n)$
\begin{equation}\label{star2000}
u(x)+\int_{B(0,n)} G_{B(0,n)}(x,y)V(y)f(u(y))dy=H(u;B(0,n))(x)\leq 1\,.\end{equation}
Let $(z_n)$ be the sequence of harmonic bounded functions defined by
$$z_n:=H(u;B(0,n))\quad\mbox{in}\ B(0,n).$$
Then $z_n\leq z_{n+1}\leq 1$ in $B(0,n)$, hence $z:\RR^N\ri\RR$ defined by $$z(x):=\sup_n z_n(x)\quad\mbox{for all $x\in\RR^N$}$$ is a harmonic bounded function.
Using the Liouville theorem we have $z\equiv C\leq 1$. Since $\sup_{\RR^N}u=1$, we deduce that
$C=1$ hence $z(x)=1$ for all $x$.

On the other hand, using the properties of the Green function we have
$$\sup_n G_{B(0,n)}(x,y)=G_{\RR^N}(x,y).$$

Passing now to the limit as $n\ri\infty$ in relation \eqref{star2000} we obtain \eqref{enunt}.
\qed

\smallskip
Conversely, we prove that if problem \eqref{usolves} admits a nonnegative nontrivial bounded solution, then the function $u$ defined by relation \eqref{udefinition} is nontrivial and it is a fixed point of the integral operator
$$Sv(x):=1-\int_{\RR^N}G(x,y)V(y)f(v(y))dy.$$
More precisely, the following property holds.

\begin{proposition}\label{propo2}
Assume that problem \eqref{usolves} admits a nonnegative nontrivial bounded solution. Then the function $u$ defined by \eqref{udefinition} satisfies relation \eqref{enunt}.
\end{proposition}

{\it Proof.} Recall that
$$u(x):=\inf\{ \tilde u_n(x);\ n\geq 1\},$$
where $u_n$ is the unique solution of the Dirichlet problem \eqref{p1n}. Using Lemma \ref{l1} we have for all $x\in B(0,n)$
$$u_n(x)+\int_{B(0,n)} G_{B(0,n)}(x,y)V(y)f(u_n(y))dy=1.$$
This shows that in order to conclude the proof, it is enough to show that
\begin{equation}\label{perugia}
\int_{B(0,n)} G_{B(0,n)}(x,y)V(y)f(u_n(y))dy\ri
\int_{\RR^N} G_{\RR^N}(x,y)V(y)f(u(y))dy\quad\mbox{as}\ n\ri\infty .\end{equation}

Let  $U$ be a bounded solution of problem \eqref{usolves}. By rescaling, we can assume without loss of generality that $1<\sup_{\RR^N}U<+\infty$.
In order to prove \eqref{perugia}, we consider the auxiliary problem
$$\left\{\begin{array}{lll}
&\Delta z_n = V (x)\, f(z_n)&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
& z_n=\sup U&\qquad \mbox{if}\
|x|=n\\
& z_n\geq 0&\qquad  \mbox{in}\
B(0,n) .\end{array}\right.$$
Using Lemma \ref{l1} we obtain for all $x\in\RR^N$
$$z_n(x)+\int_{B(0,n)}G(x,y)V(y)f(z_n(y))dy=\sup U.$$
Using now Proposition \ref{propo1} we have
$$\int_{B(0,n)}G(x,y)V(y)f(z_n(y))dy=\sup U-z_n(x)\ri
\int_{\RR^N}G(x,y)V(y)f(z(y))dy,$$
where $z:=\inf z_n$. We point out that $z\not=0$, since $z_n\geq u_n$ by the maximum principle. Using now the Lebesgue dominated convergence theorem we obtain \eqref{perugia}.
\qed

\smallskip
Combining Propositions \ref{propo1} and \ref{propo2} we obtain the main result of this paper, which establishes a characterization property for the existence of bounded solutions of problem \eqref{p1}.

\begin{theorem}\label{tmain}
Let $$\tilde u_n(x):=\left\{\begin{array}{lll}
&  u_n(x)&\qquad\mbox{if}\ |x|\leq n\\
& 1 &\qquad\mbox{if}\ |x|>n,\end{array}\right.
$$
where $u_n$ is the unique solution of the problem
$$\left\{\begin{array}{lll}
&\Delta u = V (x)\, f(u)&\qquad \mbox{in}\
{\mathcal D}'(B(0,n))\\
& u=1&\qquad \mbox{if}\
|x|=n\\
& u\geq 0&\qquad  \mbox{in}\
B(0,n) .\end{array}\right.
$$

Define
$$
u(x):=\inf\{ \tilde u_n(x);\ n\geq 1\}.$$

Then problem \eqref{p1} admits a bounded distributional solution if and only if
 $$\sup_{x\in\RR^N}u(x)=1.$$

 In such a case, $u$ satisfies the additional property
$$
u(x)+\int_{\RR^N}G(x,y)V(y)f(u(y))dy=1\quad\mbox{for all $x\in\RR^N$}.
$$
\end{theorem}

\smallskip
\indent {\bf Acknowledgements.} The authors acknowledge the technical and financial support of the Deanship of Scientific Research, King Abdulaziz University of Jeddah, through Grant No. 39-130-35-HiCi.



\end{document}
